Title: Antiphospholipid syndrome; its implication in cardiovascular diseases: a review

Authors:

Ioanna Koniari (iokoniari@yahoo.gr)
Stavros N Siminelakis (ssiminel@yahoo.com)
Nikolaos G Baikoussis (ngbaik@yahoo.com)
Georgios Papadopoulos (gpapadopoulos@hol.gr)
John Goudevenos (igoudev@cc.uoi.gr)
Efstratios Apostolakis (stratisapostolakis@yahoo.gr)

Version: 2 Date: 27 August 2010

Author’s response to reviews:

Cover letter – Journal of Cardiothoracic Surgery

Patras, August 28, 2010

To: the Editor-in-Chief
Journal of Cardiothoracic Surgery

Dear Sir or Madam,

We herewith submit the revised article entitled “Antiphospholipid syndrome; its implication in cardiovascular diseases; review”, to be considered for publication in the Journal of Cardiothoracic Surgery.

The manuscript was corrected in accordance with the Journal of Cardiothoracic Surgery reviewers. All the corrections proposed-requested by the reviewers was done.

Sincerely

I. Koniari MD